Author's response to reviews

Title: How Much Dentists are Ethically Concerned about Overtreatment; a Vignette-Based Survey in Switzerland

Authors:

Ali Kazemian (Ali.Kazemian@usb.ch)
Isabelle Berg (isabelle.berg@usb.ch)
Christina Finkel (cristinafinkel@yahoo.com)
Shahram Yazdani (shahram.yazdani@yahoo.com)
Hans-Florian Zeilhofer (hfzeilhofer@uhbs.ch)
Philipp Juergens (pjuergens@uhbs.ch)
Stella Reiter-Theil (s.reiter-theil@unibas.ch)

Version: 6 Date: 19 May 2015

Author's response to reviews:

Dear Editor:

I am pleased to resubmit for publication the revised version of MS#137137881144856" How Much Dentists are Ethically Concerned about Overtreatment; a Vignette-Based Survey in Switzerland". I appreciated the constructive criticisms of the reviewer 2. I have addressed each of his concerns as outlined below. Moreover, I would like to thank the reviewer 1 for his positive feedback.

Regarding the concerns mentioned in the items 6 (Are the discussion and conclusions well balanced and adequately supported by the data?) and 9 (Do the title and abstract accurately convey what has been found?) some modifications were made through lines 20, 21, 208 and 338.

In order to address the issue concerning the limitations of the study (7. Are limitations of the work clearly stated?), we added lines 319-322.

Please let us know if any other modification was wanted by the reviewers.

Sincerely,

Ali Kazemian
DDS. PhD.
Department of Community Oral Health, School of Dentistry, Mashhad University of Medical Sciences, Mashhad, Iran.
00989155033050
kazemiana@mums.ac.ir, Ali.Kazemian@usb.ch